Author's response to reviews

Title: Periodontal probing of an impacted tooth recovered through a surgical-orthodontic approach: a case report

Authors:

maria teresa dinoi (teresadinoi@hotmail.it)
mariano lacarbonara (mariano.lacarbonara@gmail.com)
salvatore dimartino (s.dimartinoteramo@inwind.it)
annalisa monaco (annalisamonaco@yahoo.it)
giuseppe marzo (giuseppe.marzo@cc.univaq.it)

Version: 4 Date: 11 November 2013

Author's response to reviews:

Dear Editor,

I'm sending you our manuscript entitled “Periodontal probing of an impacted tooth recovered through a surgical-orthodontic approach: a case report” by Dinoi Maria Teresa, Lacarbonara Mariano, DiMartino Salvatore, Monaco Annalisa and Giuseppe Marzo from the University of L'Aquila, Italy. We would like to have the manuscript considered for publication in "Journal of Medical Case Reports". We think it's a case report original and similar projects lately have not been presented, is original to the topic, and because it takes into account the various specialties in dentistry We think it is a case report interesting because it takes into account more discipline in dentistry, such as pediatric dentistry, orthodontics, surgery and periodontology. We confirm that this manuscript has not been published elsewhere and is not under consideration by another journal. All authors have approved the manuscript and agree with its submission to this journal. Please let me know of your decision at your earliest convenience.

With my best regards,

Maria Teresa Dinoi